Author's response to reviews

Title: Premarital Sexual Practices and its Predictors among in-School youths in Shendi Town, West Gojjam Zone, North Western Ethiopia

Authors:

Alemayehu Bogale (alemayehubog@gmail.com)
Assefa Seme (assefaseme@gmail.com)

Version:2 Date: 11 March 2014

Author's response to reviews:

This study was conducted to determine the factors associated with premarital sexual practices among in-school youths. The publication of this finding has a paramount importance on local governmental and nongovernmental organizations working on youths on sexual and reproductive health issues to design polices and strategies to prevent in-school youths from early age at sexual initiation and its related risky sexual behaviors. We have tried to revise the overall manuscript from abstract up to end based on the valid comments received from peer reviewing journals experienced reviewers and editors. We are very glad to thank them for their constructive comments and time for reviewing this article. We tried to highlight the major changes edited on the article in different color specifically on purple.